Author’s response to reviews

Title: Down-regulation of the expression of CCAAT/enhancer binding protein alpha gene (C/EBPalpha) in cervical squamous cell carcinoma

Authors:

Zemin Pan (panteacher89@sina.com)
Weinan Zheng (rwz77666@163.com)
Jinli Zhang (zhangjinli@shzu.edu.cn)
Rui Gao (gaorui_shz@sina.com)
Dongmei Li (lidong_abc@126.com)
Xiaoqing Guo (gxqj66@sina.com)
Hu Han (49166070@qq.com)
Feng Li (lifeng7855@yahoo.com.cn)
Shen Qu (qushen@mails.tjmu.edu.cn)
Renfu Shao (rshao@usc.edu.au)

Version: 2 Date: 12 November 2013

Author’s response to reviews: see over
Dear editor,

I submit my manuscript in your journal. I am the corresponding author. I declare information below for this manuscript.
A. This manuscript has not been published nor submitted for publication elsewhere.
B. All authors have contributed significantly, and that all authors are in agreement with the content of the manuscript.
C. This paper a new finding is that C/EBPα protein level was significant difference between chronic cervicitis and cervical carcinoma. C/EBPα mRNA level was significantly lower in cervical cancer tissues than in normal cervical tissues. Methylation of promoter of C/EBPα gene was significant higher in cervical cancer than in normal cervical tissues. CEBPα pcDNA3.1 construct transfected into HeLa cells inhibited cell proliferation and decreased cell migration. All of these found are original unpublished work.
D. Contacted corresponding author is Zemin Pan.

**Mailing Address:** Department of Biochemistry and Molecular Biology, School of Medicine, Shihezi University, Shihezi, Xinjiang, 832002, China

phone: 86 993 2057882; fax: 86 993 2057882
E-mail: panteacher89@sina.com

Best regards,
Zemin Pan